Author's response to reviews

Title: The accuracy of FAST in relation to grade of solid organ injuries: A retrospective analysis of 226 trauma patients with liver or splenic lesion

Authors:

Beat Schnüriger (beat.schnueriger@insel.ch)
Joachim Kilz (joachimkilz@exadaktylos.ch)
Daniel Inderbitzin (daniel.inderbitzin@insel.ch)
Miranda Schafer (miranda.schafer@gmail.com)
Ralph Kickuth (ralph.kickuth@insel.ch)
Martin Luginbuehl (martin.luginbuehl@insel.ch)
Daniel Candinas (daniel.candinas@insel.ch)
Aristomenis Exadaktylos (aristomenis@exadaktylos.ch)
Heinz Zimmermann (heinz.zimmermann@insel.ch)

Version: 6 Date: 15 March 2009

Author's response to reviews: see over
Letter to the editor

Thank you very much for accepting our paper for publication in BMC Med Imaging. The title is changed and the limitations were added at the end of the discussion section.

Yours sincerely,

Beat Schnüriger.